Author’s response to reviews

Title: Pancreatitis in Pre-Adolescent Children: a 10 year experience in the Pediatric Emergency Department

Authors:
Melanie Randall (mrandall@llu.edu)
Sarah McDaniels (smcdaniels@llu.edu)
Kristina Kyle (kkyle@llu.edu)
Meina Michael (mmichael@llu.edu)
Julia Giacopuzzi (jgiacopuzzi@llu.edu)
Lance Brown (labrown@llu.edu)

Version: 2 Date: 16 Oct 2019

Author’s response to reviews:

Please move the Declarations section before References section. = the declarations section has been moved to before References

Please rename 'Data Availability' into 'Availability of data and materials'. = this has been changed

Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate. = this has been included